HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY COMP 102 : Pr ogramming & Computer Fundamentals I Summer 1996 Laboratory Assignment 6 5 & 7 August

Given the following definitions for Date, Student, Classlist and Class102 :

struct Date {

int day; // day of a Date ASSUME: 1 <= day <= 31 int month; // month of a Date ASSUME: 1 <= month <= 12 int year; // year of a Date ASSUME: 1 <= year <= 2000 } ; struct Student { char name[20]; // Student’s name Date birthdate; // Date of a student’s birthday int score[5] // Student’s test scores }; struct Classlist { Student clist[50]; // Current list of students in the class int no_tests; // Number of tests students have taken int size; // Number of students currently enrolled }; Classlist Class102;

Write an integer function age that takes three arguments, today’s date (of type Date), name of a student (a character string up to 19 characters long) and a class list (of type Classlist),and returns the age of the student (i.e. int age ( Date, char name[], Classlist ) ).

Write a void function withdraw that takes two arguments, name of a student (a character string up to 19 characters long) and the current class list (of type Classlist), and returns an updated class list (of type Classlist) with the student’s record deleted from the original class list (i.e. void withdraw ( char name[], Classlist& ) ).

Write a double function average that takes two argument, the test number (an integer) and the current class list (of type Classlist), and returns the average of the test scores for that given test. (i.e. double average ( int testn, Classlist ) ).

Write a program that asks the user to input the number of tests and the number of students and reads a Classlist from the file “Class102.dat” into Class102 using a function in_class whch will be provided in class. It will then allow the user to select repeatedly one of the following operations until a -1 is entered:

1. Ask for the name of a student and return the age of the student. 2. Ask for the name of a student and delete the student’s record from Class102. 3. Compute the average of the test scores for a given test in the current Class102.